Ops 1/23

Another beautiful day in the Heber valley. Clear and cold this morning (15 degrees when we left the hotel!) but warmed up and got hazier in the afternoon.
This morning, after some rass troubleshooting at iss1, we went to DCS while it was frozen enough to drive in, so | could bring the ladder and get the last
couple EC100 serial numbers. We also checked out the tsoil that had stopped reporting yesterday-visually everything was fine, after unplugging and
replugging the binder connections it is again. Glad we didn't have to try burying another one in the frozen ground.

Spent the afternoon showing Liz how I've been checking data for ISFS. Matt is arriving this evening to take over from me.

Heard from the Pls that there will not be an IOP starting tomorrow, so replacing the 32m thermocouple at PRS is not urgent, but will need to happen before
the next IOP.

No new data issues today. Reset some HRXLs (sp, dcs, cc—the usual ones), noticed at one point that prsr was off the net but it came back successfully on
its own.
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